Author's response to reviews

Title: Genetic association study of selected candidate genes (APOB, LPL, LEPTIN) and telomere attrition in obese and hypertensive individuals.

Authors:

Birajalaxmi Das (birajalaxmi@yahoo.co.in)
Nilambari Pawar (nilambari.pawar@gmail.com)
Divyalakshmi Saini (ndivya@barc.gov.in)
M Seshadri (msesh@barc.gov.in)

Version: 2 Date: 15 December 2008

Author's response to reviews:

To
Editor-in-Chief,
BMC Medical Genetics,

Dear Sir/Madam,

I am submitting the manuscript entitled, “GENETIC ASSOCIATION STUDY OF SELECTED CANDIDATE GENES (APOB, LPL, LEPTIN) AND TELOMERE ATTRITION IN OBESE AND HYPERTENSIVE INDIVIDUALS.” Authored by Birajalaxmi Das, Nilambari Pawar, Divyalakshmi Saini and M. Seshadri to your esteemed journal.

Awaiting for a positive reply.

Thanking you,

Dr. Birajalaxmi Das,
LLRSS, RB&HSD,
BARC, Trombay,
Mumbai 400 085.
India.
birajalaxmi@yahoo.co.in